Author’s response to reviews

Title: Video-assisted Thoracoscopic Cardiac Denervation of Refractory Ventricular Arrhythmias and Electrical Storms: A Single-center Series at the Cardioinfantil Foundation

Authors:
Luis Téllez (luisjaime@me.com)
Juan Garzón (jgarzon2003@me.com)
ERIC VINCK (evinck518@gmail.com)
Julian Castellanos (jdcastellanosg@unal.edu.co)

Version: 1 Date: 02 Oct 2018

Author’s response to reviews:

Thank you very much for your reviews, we revised the recommended areas of concern and had an English speaking physician take a look at our article. He made the recommended modifications.

Note that the "arrhythmia and electrophysiology" is one department.